

Working Paper Series 7

Explaining Bank Failures in Brazil: Micro, Macro and Contagion Effects (1994-1998)

Adriana Soares Sales and Maria Eduarda Tannuri-Pianto October, 2007

ISSN 1518-3548 CGC 00.038.166/0001-05

Working Paper Series

Edited by Research Department (Depep) - E-mail: workingpaper@bcb.gov.br

Editor: Benjamin Miranda Tabak – E-mail: benjamin.tabak@bcb.gov.br Editorial Assistent: Jane Sofia Moita – E-mail: jane.sofia@bcb.gov.br Head of Research Department: Carlos Hamilton Vasconcelos Araújo – E-mail: carlos.araujo@bcb.gov.br

The Banco Central do Brasil Working Papers are all evaluated in double blind referee process.

Reproduction is permitted only if source is stated as follows: Working Paper n. 147.

Authorized by Mário Mesquita, Deputy Governor for Economic Policy.

General Control of Publications

Banco Central do Brasil Secre/Surel/Dimep SBS – Quadra 3 – Bloco B – Edifício-Sede – 1° andar Caixa Postal 8.670 70074-900 Brasília – DF – Brazil Phones: (5561) 3414-3710 and 3414-3567 Fax: (5561) 3414-3626 E-mail: editor@bcb.gov.br

The views expressed in this work are those of the authors and do not necessarily reflect those of the Banco Central or its members.

Although these Working Papers often represent preliminary work, citation of source is required when used or reproduced.

As opiniões expressas neste trabalho são exclusivamente do(s) autor(es) e não refletem, necessariamente, a visão do Banco Central do Brasil.

Ainda que este artigo represente trabalho preliminar, citação da fonte é requerida mesmo quando reproduzido parcialmente.

Consumer Complaints and Public Enquiries Center

Address:	Secre/Surel/Diate
	Edifício-Sede – 2º subsolo
	SBS – Quadra 3 – Zona Central
	70074-900 Brasília – DF – Brazil
Fax:	(5561) 3414-2553
Internet:	http://www.bcb.gov.br/?english

Explaining Bank Failures in Brazil: Micro, Macro and Contagion Effects (1994-1998)*

Adriana Soares Sales[†] Maria Eduarda Tannuri-Pianto[‡]

The Working Papers should not be reported as representing the views of the Banco Central do Brasil. The views expressed in the papers are those of the author(s) and do not necessarily reflect those of the Banco Central do Brasil.

Abstract

We apply duration (survival) models with exponential and exponential piecewise-constant hazard functions to a panel of 273 banks to study the determinants of bank failure over the 1994-1998 period in Brazil. The models deal empirically with left censoring in the data. We control for macroeconomic conditions and contagion effects, as well as bank-specific factors. Our results indicate that foreign banks have distinct empirical survival functions relative to other banks. For Brazil, macroeconomic and bank-level covariates explain the likelihood and timing of bank failure. Our indicator of system-wide financial fragility (IFF) suggests that the banking industry faced increased fragility after November 1995. We find (some) evidence that the Program of Incentives for the Restructuring and Strengthening of the National Financial System (Proer) was able to distinguish solvent from insolvent banks.

Keywords: Duration Analysis; Exponential Hazard; Bank Failures; Brazil. JEL Classification Codes: C41 and G21.

^{*}The authors acknowledge and appreciate the comments of Fabiana Rocha and Rodrigo Peñaloza. All non-reported results can be obtained from the authors upon request.

[†]Banco Central do Brasil. E-mail: adriana.sales@bcb.gov.br.

[‡]Universidade de Brasília. E-mail: tannuri@unb.br.

1 Introduction

From the beginning of the Real Plan in July 1994 until December 1998, 83 banks – commercial banks, universal banks and savings and loans associations – suffered some type of intervention, including from the adoption of special regimes by the Central Bank of Brazil (BC) to any type of stockholder restructuring (mergers, incorporations, cancellations, changes of social object and split-ups) or privatizations. It is worth mentioning that 59 banks "failed" during the period, which represents around 22% of the sample of all banks alive.

That "mortality" pattern can be studied using duration analysis, which models a Markov process with two states - "alive and dead." These models are able to explain and predict how the conditional probability of failure evolves over time, which covariates affect it and when the potential failure occurs.

In this paper, we investigate the determinants of bank failure and bank unsoundness in Brazil over the 1994-1998 period. We apply duration (survival) analysis to a panel of "time-to-event" data, which allows disentangling bank - specific factors from others that lead to bank weakness. Since banking crises can be costly¹ it seems crucial that the financial system regulator collects information about the sources of banking fragility to institute remedial action at troubled banks.

We also develop an indicator of system-wide financial fragility $(IFF)^2$, which is useful for financial surveillance. The estimated models enable us to estimate the mean survival time of each bank in each month of the 1994-1998 period. It also provides evidence that the Program of Incentives for the Restructuring and Strengthening of the National Financial System (Proer) was able to distinguish solvent from insolvent banks and avoid contagion.

The main methodological contributions of the paper are: i) dealing with left cen-

¹A report by the World Bank (1999) mentions an IMF study that calculates the average cost of a banking crisis in terms of a GDP loss as 14.6% (over a trend) per crisis. See also Bernanke and Gertler (1990).

 $^{^{2}}$ An economy exhibits financial fragility if it possesses a propagation mechanism that allows even small exogenous shocks to generate crises with relevant effects on the financial structure and on the real activity of the economy.

soring in the data; ii) incorporating time-varying covariates; and iii) controlling for bank-specific covariates, macroeconomic conditions, and potential contagion effects.

Our results indicate that: i) macroeconomic conditions contribute to explain bank failures and improve in and out-of-sample accuracy of the models; ii) foreign banks have a different survival function and should be analyzed disjointly; iii) there is no relevant bank-level liquidity indicator; iv) panel data information should be used in these studies; and v) there is evidence that Proer avoided deposit runs.

The paper is organized as follows. Section 2 provides some background on and discussion of the main issues in this literature. Section 3 briefly describes the Brazilian Banking system over the 1994-1998 period. Section 4 presents the data and the expected effects of the covariates on bank fragility. Section 5 develops the methodology. Section 6 discusses the results. Some illustrations of models use are presented in Section 7. Section 8 concludes.

2 Background

The literature that uses duration models to explain and predict bank failures for the US Banking system includes Lane, Looney and Wansley (1986) who use a nonparametric proportional hazard model introduced by Cox (1972) and non-time varying bank-specific covariates for banks that failed between January 1978 and June 1984. DeYoung (2000) employs a split-population duration model with log-logistic hazard function and time-varying bank-level covariates as well as regulatory and economic environment variables to examine the financial performance of commercial banks chartered between 1980 and 1985. Wheellock and Wilson (2000) model failures and acquisitions using a competing-risk hazard model framework. They utilize the Cox model with time-varying bank level covariates over the 1984-1993 period.

For the Argentinean Banking system, Dabós and Escudero (2000) use the Cox model with non-time varying bank-specific covariates (from November 1994). Bercoff et al. (2002) employ a lognormal hazard function and time-varying macroeconomic covariates and (non-time-varying) bank-specific covariates over the 1993-1996 period. For the Mexican Banking system González-Hemossillo et al. (1997) estimate a split-population survival model with a log-logistic hazard function and bank-specific, macroeconomic and contagion variables from 1991 to 1995. Only the macroeconomic variables are time-varying.³ For the Brazilian Banking system Rocha (1999) and Janot (2001) employ the Cox model and non-time-varying covariates. The former considers banks that failed between July 1994 and December 1995 and the latter, between 1995 and 1996 (covariates from December 1994).

Usually, when duration models are used to study bank failure (for example Lane et al., 1986; González-Hermosillo, 1999) the initial time period of the analysis does not coincide with the true period in which banks are born. This practice results in left censoring of the data, which is not taken into account during the statistical estimation procedure. In this article we propose a hazard function that overcomes this left censoring problem, which is the exponential function, a "memoryless" hazard.

González-Hermosillo (1999) discusses the "life cycle of bank failures" and the importance of capturing the "whole cycle". In order to accomplish this task the author suggests employing time-varying covariates. We adopt this approach in our analysis.

The simultaneous use of bank-level indicators and macroeconomic variables proves to be very important. The macroeconomic approach of bank failure cannot independently explain why some banks survive macroeconomic shocks and others fail. On the other hand, the bank-level approach ceases to be efficient if one does not account for varying economic conditions. In this sense, we include both types of variables in our models. Furthermore, we follow Bercoff *et al.* (2000), González-Hermosillo (1999), González-Hermossilo *et al.* (1997) and D'Amato *et al.* (1997) when choosing contagion proxies.

³It is not clear if the bank-specific covariates are time-varying as well.

3 The Brazilian Banking System and Proer

Before the implementation of the Real Plan in July 1994 the Brazilian banking system was based on high-inflation rate funds transfers from the depositors to the banks. Despite accounting for a small portion of the banks' liabilities, the demand deposits bore a strong negative *ex-post* real interest rate. If the "Ponzi-game condition" applies, which means here that the bank's deposits growth rate is higher than the interest rate paid for the bank's liabilities, then the system functions well and is likely over-dimensioned.

After the Real Plan, inflationary transfers to the banking system decreased by almost R\$ 9 billion until 1996⁴ (around USD 4.1 billion). It is possible that in some moment after the Plan the bank's deposits growth rate was lower than the interest rate paid by the banks to the depositors. Under this situation the banks have basically two choices: make a fund transfer from the borrowers or sell their stock of net assets. If the economy suffers a shock, as during the Mexican crisis in December 1994, and the transfers reach very high levels, the system can face a banking crisis.

During this period, the loss of floating revenue was compensated by a rise in loan revenues⁵, which generated a "lending boom". There is evidence that both in industrialized and Latin American countries such booms are sometimes followed by banking crises: it becomes more difficult to disentangle good from bad borrowers. This is reinforced by the evolution of the credit default rate after the Real Plan: to consumers it increased to 16.67% of the total credit operations from July to November 1995 from 4.38% in the second semester of 1994.

Therefore, after the Plano Real, the Central Bank of Brazil had to regulate an over-dimensioned banking system. As part of the main reforms were the Basle Capital Agreement in 1994, and the inclusion of many state-owned banks such as

 $^{^{4}}$ See Cysne (1997).

⁵In such a situation, it seems that the banks are not willing to reduce costs because any bank that delays the adjustment can increase its market share.

Banespa and Banerj in the Special Regime of Temporary Administration (Raet). In spite of several preventive measures, a banking crisis seemed to have settled in after the intervention of Banco Econômico in August 1995.

To avoid the high costs of a system-wide banking crisis, the Central Bank implemented the Program of Incentives for the Restructuring and Strengthening of the National Financial System (Proer) in November 1995. The credit lines and the subsidized tax treatment granted under Proer were aimed at promoting bank mergers and incorporations by other institutions. Seven banks were either merged, incorporated or had their shareholder control transferred while making use of Proer resources. Under the Proer, the Central Bank restructured 4 private banks (Econômico, Bamerindus, Nacional and Banorte) using the "good bank/ bad bank" failure resolution method and it granted credit lines to the acquirer banks, as Table I indicates.

Regime		Operation	Amount millions ^{1/}	in	R\$
Banks	under	Sale of part of assets and	19,108		
Intervention		liabilities to:			
Econômico		Banco Excel e Caixa Econômica			
		Federal			
Nacional		Banco Unibanco			
Mercantil de PE		Banco Rural			
Banorte		Banco Bandeirantes e CEF			
Bamerindus		HSBC e CEF			
Banks not	under	Transfer of Stockholder Control	1,251		
Intervention		to:			
United		Banco Antônio Queiroz			
Martinelli		Banco Pontual			

Table I: Sale of Banks with Resources from Proer

Source: Central Bank of Brazil.

¹⁷ Historical values.

4 Data

Our initial sample consists of all 273 banks that were operating in July 1994 or opened from July 1994 through December 1998 (24 banks). The covariates are time-varying from 1994 to 1998. The sample does not include Banco Morgan Stanley Dean Witter, Banco Holandês Unido, Banco Paraiban, Banco Banerj and Banco Pottencial, for which complete data was not available. The bank-level series were discontinued in 1999. The sample period is homogeneous in terms of regulation, the technical production possibilities of the banking industry, and economic environment.

In this paper, we define a bank survival spell as the length of time that a bank stays solvent. As the events of bank unsoundness and legal bankruptcy do not necessarily occur at the same time, we consider as insolvent banks that have undergone Special Regimes - which include Raet, intervention and out-of-court liquidation have been canceled, or have received financial support from the Central Bank of Brazil during the sample period.

We also consider banks that are cancelled due to change of business goals, incorporations, transfers of stockholder controls, and privatizations to have a censored lifetime, with censor date equal to the date of the event. Incorporated banks are treated like merged banks since it is not possible to split them into separated banks. Therefore, both the incorporated banks and the incorporating banks have a censored duration. Banks that suffer transfers of stockholder control operate separately and only the acquired bank has a censored lifetime.

Event	Indicator in the date of the event
Raet, Intervention and	
Liquidation	1
Proer, Proes	1
Cancellation	1
Cancellation due to change	
of business goals	0
Incorporation and Transfer	0
of Stockholder Control	
Privatization	0

Describuon or Events

Obs: 0 =censoring e 1 =failure.

Table II summarizes the breakdown of failure events in our sample.

Table II: Summary	of Failure Events
--------------------------	-------------------

Event	Number
Raet	5
Intervention	7
Liquidation	32
Proer, Proes	5
Cancellation	10
Total	59

Source: PCIF 400 from Sisbacen.

Note: The first event that occurred since July 1994.

We assume that bank stratification follows the equity capital ownership criteria: federal and state-owned banks, private national banks, foreign banks and private national banks with foreign capital, named types 1, 2, 3 and 4 respectively. Table III describes the number of failures and observed duration by type of bank.

	N⁰ of	N⁰ of				
Туре	banks	failures	Observed duration (in months) ^{2/}			
			Minimum	Mean	Maximum	
1- State-owned	30	9	5	43.8	53	
2- Private National ^{1/}	163	44	2	41.2	53	
3- Foreign-owned	47	1	7	46.7	53	
4- Priv. Nat. w/Foreign	31	5	21	49.0	53	
Total	271	59				

 Table III: Number of Failures and Duration by Type

^{1/} The Bancos Hércules and Garavelo were excluded.

^{2/} The duration is calculated over the 1994-1998 period.

Our hypothesis here is that bank fragility is determined by bank-specific factors, macroeconomic conditions, and contagion effects. The covariates in Appendix A.1 control for those aspects. The Appendix A.1 also shows the expected effect of the variables (signs in the statistical models) on the estimated survival times. Our data source is the Central Bank of Brazil.

We select and test 29 out of 68 bank-level indicators from the "System of Institutions under Attendance and Control of the Central Bank of Brazil", hereafter called INDCON system, collected every semester. The INDCON system classifies the indicators into five groups representing: capital adequacy (C), asset quality (A), profitability (R), efficiency (E), and liquidity adequacy (L).⁶

We include several forward and backward-looking macroeconomic variables, available either per month or semester. Besides those variables described in the Appendix A.1 we tested the Selic nominal rate, the General Price Index-Market (IGP-M) and the spread over Treasury for the C-BOND. All macroeconomic covariates were tested under several lags and frequencies. The typical included variable varies monthly and is lagged for three-months. The variables leaded by 0, 2 and 6 remain constant over each six-month interval. For example, the ones followed by 0 are the variables in June and in December of each year from 1994 to 1998; the ones followed by 2 have a two-month lag and by 6 a six-months lag. To have some sense of the initial lags, we use an event study methodology, which consists in examining the variable in level and its mean until twelve months prior to the failure event and comparing it to the variable mean outside the twelve-month window.

We choose contagion variables that affect the banking system as a whole and have been related to herding behavior and deposits run. Economies of scale, portfolio diversification, and bank runs *a la* Diamong-Dybvig (1983) are addressed by specific covariates (see Appendix A.1). Economies of scale and portfolio diversification effects are represented by the variable *ativoreal*, which represents the asset value in the current month. For the variables *ativorealinício*, *ativorealmeio* and *ativorealfim* we make the asset value constant over the six-month interval, but employ the values at the beginning, middle and end of the half-year period, respectively.

Bank runs are addressed by the demand deposits growth rate. To avoid endogeneity problems, we instrument the variable by using the demand deposits growth rate lagged three months (*vardvlag3*). The variable *vardvmean* is the average of the demand deposits growth rate over the six-month interval.⁷

⁶The INDCON system follows the US "Uniform Financial Institutions Rating System (UFIRS)" also called the "CAMEL rating system". Under the CAMEL rating system each financial institution is assigned a composite rating based on an evaluation of components that address the capital adequacy, the assets quality, the capability of management, the quality and level of earnings, liquidity, and the sensitivity to market risk.

⁷The results including demand deposits growth rate are not reported.

Concerned about moral hazard problems, we assess whether the establishment of a formal scheme of insurance deposit in Brazil has changed the probability of bank failure. The dummy variable fgc (1 after December 1995 and 0 otherwise) was not statistically significant in any statistical estimation and was excluded.

5 Methodology

Duration models are usually applied to censored data, and therefore one should derive a likelihood function for such data. For a random draw i from the population, let t_i denote the time for which individual i is observed and let t_{i^*} denote the duration. If i dies with $t_i < t_{i^*}$, there is no right censoring and estimation employs traditional conditional maximum likelihood (CLE). In order to account for right censoring, we assume that the observed duration t_i is obtained as $t_i = \min(t_{i^*}, b_i)$, where b_i is the censoring time for individual i. The probability that t_i is censored is:

$$P(t_i^* > bi \mid x_i) = 1 - F(b_i \mid x_i; \theta)$$

where F(.) is the conditional cdf of t_{i^*} given x_i - the vector of observed covariates and θ - the vector of unknown parameters.

Let d_i denote a binary failure indicator that equals one if a bank fails and zero otherwise and f(.) the probability density function. The conditional likelihood function for the N-size sample is:

$$L = \prod_{i=1}^{n} L_{i} = \prod_{i=1}^{n} f(t_{i} \mid x_{i}; \theta)^{d_{i}} [1 - F(t_{i} \mid x_{i}; \theta)]^{1-d_{i}}$$
$$= \prod_{i=1}^{n} \lambda(t_{i} \mid x_{i}; \theta)^{d_{i}} S(t_{i} \mid x_{i}; \theta)^{3}$$
(1)

where

$$\lambda(t;x) = \lim_{\Delta t \downarrow 0} \frac{P(t \le T \le t + \Delta t \mid T > t, x)}{\Delta t}$$

and

$$S(t;x) = \exp\left[-\int_{0}^{t} \lambda(s;x)ds\right] = \exp\{-\Lambda(t;x)\}$$

are the hazard and the survival functions.

Using the integrated hazard function $\Lambda(t_i \mid x_i; \theta)$, we see that:

$$\log L = \sum_{i=1}^{n} \{ d_i \log[f(t_i \mid x_i; \theta)] + (1 - d_i) \log[1 - F(t_i \mid x_i; \theta)] \}$$

=
$$\sum_{i=1}^{n} \{ d_i \log \lambda(t_i \mid x_i; \theta) - \Lambda(t_i \mid x_i; \theta) \}$$
(2)

For the grouped data, we divide the time line into J+1 intervals $(0, \tau_1], (\tau_1, \tau_2], ..., (\tau_{J-1}, \tau_J], (\tau_J, \infty)$. Let c_j be a binary censorship indicator, which is equals to one if the duration is censored in interval j, and zero otherwise, and y_j be a binary indicator equal to one if the duration ends in the j-th interval and zero otherwise. Note that $c_j = 1$ implies $c_{j+1} = 1$ and $c_{J+1} \equiv 1$. Also, $y_j = 1$ implies $y_{J+1} = 1$. If $c_j = 1$, we adopt by convention that $y_j \equiv 1$.

Following Wooldridge (2002), only two combinations of y_j , y_{j-1} and c_j yield probabilities that are not identically equal to zero or one, which are:

$$P(y_j = 1 \mid y_{j-1} = 0, x, c_j = 0) = 1 - \exp\left[-\int_{\tau_{j-1}}^{\tau_j} \lambda(s; x, \theta) ds\right]$$
$$\equiv 1 - \alpha_j(x, \theta)$$

and

$$P(y_j = 0 \mid y_{j-1} = 0, x, c_j = 0) = \alpha_j(x, \theta)$$

Then, if for observation i, an uncensored exit occurs in j_i , the likelihood is:

$$L_{i} = \left[\prod_{h=1}^{j_{i}-1} \alpha_{h}(x_{i}, \theta)\right] \left[1 - \alpha_{j}(x_{i}, \theta)\right]$$
(3)

where $\alpha_j(x,\theta) \equiv \exp\left[-\int_{\tau_{j-1}}^{\tau_j} \lambda(s;x,\theta) ds\right].$

If the duration is censored in interval j_i , the likelihood is only the first term on the right hand side of equation (3).

In the case of time-varying covariates that are constant within each time interval, the (partial) likelihood is:

$$L_i = \left[\prod_{h=1}^{ji-1} \alpha_h(x_{ih}, \theta)\right] \left[1 - \alpha_{j_i}(x_{i,j_i}, \theta)\right]$$
(3')

In the paper, we implement CLE (Conditional Likelihood Estimation) using exponential and piecewise-constant exponential (PCE) hazard functions in eq. (3').

6 Results

We first present (Figure 1) baseline estimations of the unconditional survival functions (Kaplan-Meier estimator) for the true spells of the banks (considering the effective date they started business) and for spells beginning in July 1994 (left censored data).

The two curves have very distinct behavior, with the one on the left a lot steeper than the one on the right (left censored data). This is an indication that ignoring the left censoring may distort the survival function estimations.

In order to investigate the issue of continuous *vis-a-vis* discrete data, the survival functions were also estimated using daily instead of monthly data. They behave very similarly, indicating that they do not depend on the level of aggregation.

We performed non-parametric rank tests (of survival time) for the equality of survival functions among the four types of banks considered. Visual inspection of the survival functions indicated that foreign banks had a distinct behavior. We use the log-rank, the Fleming-Harrington and the Peto-Peto-Prentice tests (Neumann, 1997).



Figure 1: Empirical Survival Functions - Monthly data

The log-rank test is more appropriate under the assumption that the hazard functions are proportional among groups, if they are not the same. The Fleming - Harrington is a generalization of the first one but we can give different weights to the events, depending on the point in time they happen. Because one of the applications of our paper is to verify the presence of contagion, we give higher weights to the events that occur more towards the beginning of the analyzed period, when the Proer was implemented. The Peto-Peto-Prentice test is appropriate when we assume that the hazard functions are affected by differences in the censoring structure among the banks. All three tests reject the null hypothesis that all four types of banks have the same survival function. However, the null cannot be rejected when comparing only state-owned (type 1), private national (type 2) and private national with foreign capital (type 3) banks, as you can see in Appendix A.2 to the log-rank test. We obtain similar results when considering only the banks which collect demand deposits.

Table IV shows average survival times by type of bank, using data with the actual birth dates of the banks and the left censored data (starting in July 1994). The restricted mean is the area below the KM survival function from July 1994 to

December 1998. The extended mean is obtained by taking the KM survival function to zero using an exponential curve, and then computing the total area under the curve.

		Restricte	ed Mean	Extende	d Mean
		Actual Birth Beginning		Actual Birth	Beginning
Туре	Number	Date	July 1994	Date	July 1994
State-owned	30	967.18	44.20(*)	967.18	146.53
Private National	165	859.32	44.93(*)	859.32	152.02
Foreign-owned	47	1601.86(*)	52.95(*)	59946.92	1935.04
Priv. Nat. w/Foreign	31	708.08(*)	50.56(*)	2668.89	288.91
total	273	1003.83	46.86(*)	1003.83	196.69

Table IV: Survival Time - Restricted and Extended Means (in months)

(*) The greatest time observed in our analysis is censored, indicating that the average is underestimated.

The previous results corroborate the idea that foreign-owned banks have a distinct survival behavior, and therefore should not be analyzed together with the other type of banks. Also, the issue of left censored data should not be ignored. If we consider the actual birth date, the state-owned banks seem to have a higher survival time than the private national ones, and this result disappears if we censor our data in July 1994.

In the Appendix A.2 we can see several microeconomics variables the have different means (t-statistic) among survivors and failed banks. The upper side of the Table "Descriptive Statistics" shows the variables employed in the models and the lower side those rejected.

6.1 Conditional Models

Because we are concerned with the colinearity problem that might be present in these kinds of models, we start our model selection process by performing a factorial analysis on the data in order to keep the variables strongly associated with the main factors and achieve an initial reduction in the number of bank-level covariates.⁸ As you can see in Appendix A.2, the factorial analysis retains 11 factors. The proportion of the total variance explained by the factors is low. Therefore, we further use

⁸The results for the factorial analysis are available from the authors upon request.

the explanatory and forecasting accuracy of the variables as a selection criterion. All estimates use a robust var-cov matrix based on efficient-score residuals. The observations are likely non-independent, given that we are dealing with covariates that vary in time.

We estimate four model specifications – A, B, C and D. Model A is an exponential model with constant hazard baseline function. It includes bank-level indicators (micro) and macroeconomic variables (IPI and Selic⁹). Model B is a piecewiseexponential model (with bank-level variabels), in which the hazard baseline functions vary in each interval considered (0,6], (6,12], (12,18], (18,24], (24,30], (30,36], (36,42], (42,48], (48,53].¹⁰ Model C is an exponential model that differs from model A in the macroeconomic and contagion variables included: Mres – ratio of total imports to international reserves (liquidity concept), and Vacre – monthly percentage change of loans.¹¹ Finally, in model D we include only microeconomic variables for comparison with the previous literature.

The microeconomic variables which are significant in all estimated models are: 1) recovery of the administrative expenses through service's income (r205), an indicator of efficiency that has a negative sign decreasing the conditional probability of failure; 2) ratio of atypical assets to total assets (a103), which indicates fraud risk and it impacts positively the probability of failure; 3) operational margin (monthly average in a semester) (r305), with a positive sign, meaning that a higher spread increases the chances of failure; 4) leverage ratio (c204) and 5) ratio of non-performing loans to total loan (a201) both are indicators of credit risk and have a positive sign; 6) loan reserve coverage which reduces the probability of failure and works as a buffer to absorb shocks; 7) ratio of other liabilities to liabilities (c107) that is a measure of a possible deterioration of the bank situation and has a positive impact on failure.

The variable general solvency (1104) is not significant in any of the regressions, a result that is not surprising, given that this variable only shows a mismatch between

⁹See the Appendix A.1 for a description of the variables and expected effects.

¹⁰The intervals are measured in months starting in July 1994 and ending in December 1998. The estimates are robust to the length of the intervals.

¹¹See the Appendix A.1 for a description of the variables and expected effects.

assets and liabilities. None of the liquidity variables are significant either, which also was not surprising given that the liquidity indicators from the INDCON system do not seem to measure liquidity well. The asset variable, which would be a measure of economies of scale and portfolio diversification, is not significant. In addition, we include a variable for the bank age at the beginning of the estimation period and it is not significant in any model, evidencing that the exponential specification is actually reasonable.

Table V shows the estimates for models A and D. We report the hazard ratio coefficients which measure the proportional effect in the hazard function from absolute changes in the covariates. The variables IPI (industrial production indicator) and Selic rate¹² are lagged for 6 months in model A. They both have positive signs, increasing the probability of a bank failure. A high interest rate increases the vulnerability of banks to shocks, while an increase in IPI could lead to lending booms, associated with economic and credit growth, and also to higher vulnerability. The sign and magnitude of the coefficients in model D are similar to the other estimated models, but its predictive power is lower as will be shown later. The main difference with the previous models is that the variable "evolution of typical assets operations" is only significant at 10%. It could be indicating lending booms and therefore would be associated with production levels and interest rate. However, it has a negative impact on the conditional probability of bank failure^{13, 14}.

¹²The Special System of Custody and Settlement of Federal Securities overnight rate (Selic), expressed in annual terms, is the average rate weighted by the volume of one-day operations backed by federal government securities, carried out at Selic system through repurchase agreements.

¹³Our results are compatible with Rocha (1999). Using a non-time varying dataset, she finds that "leverage ratio", "ratio of non-performing loans to total loans", "operational margin", and "ratio of the total funding to typical assets" are significant. In our model specification D, including only banks that collect demand deposits, all but "ratio of the total funding to typical assets" are significant as well. However, the results of Janot (2001) are substantially different, given that he finds the "ratio of liquid assets to typical assets", the "ratio of loans funded by the foreign market to liabilities", and the "ratio of administrative cost to adjusted total assets" as significant.

¹⁴Even though the variables "evolution of typical assets operations (e202)", "ratio of administrative cost to average assets (r409)" and "return on adjusted total assets (r206)" are not significant, we keep them for comparison with other studies, but our results are not affected by their exclusion.

	Model A			Model D		
	Hazard	s.e.		Hazard	s.e.	
Variables	ratio.	Robust	P>Izl	ratio	Robust	P>IzI
IPI-lag6**	1.0375	0.0172	0.027	-	-	
Selic-lag6**	1.0014	0.0006	0.024	-	-	-
Recovery Adm. Expenses**	0.9899	0.0045	0.025	0.9904	0.0038	0.013
Atypical assets/ total assets ***	1.0334	0.0078	0.000	1.0349	0.0080	0.000
Operational Margin***	1.0005	0.0001	0.000	1.0005	0.0001	0.000
Leverage ratio***	1.0002	0.0001	0.000	1.0002	0.0001	0.000
Non-performing/ total loans ***	1.0318	0.0053	0.000	1.0333	0.0051	0.000
Evol. typical assets operat. *1/	0.9999	0.0001	0.243	0.9999	0.0001	0.086
Loan reserve coverage***	0.9677	0.0085	0.000	0.9651	0.0087	0.000
Adm. Cost/ average assets	1.0752	0.0878	0.374	1.0639	0.0966	0.495
Other liabilities/ liabilities.***	1.0003	0.0001	0.000	1.0003	0.0001	0.000
Return on adjusted total assets	0.9798	0.0178	0.262	0.9844	0.0205	0.453
Real assets	1.0000	0.0000	0.185	1.0000	0.0000	0.205

Table V: Summary of Results for Models A (exponential model, bank-level and macroeconomic variables) and D (exponential model and bank-level variables)

Note: ***, ** and * indicates significance at 1%, 5% and 10% respectively. ¹Significant at 10% only for Model D.

The results for models B (PCE model with bank-level variables) and C (exponential model, contagion, macro and bank-level variables) are in Appendix A.2, and there are some interesting findings. The coefficients of the microeconomic variables in the piecewise-exponential model B have signs and magnitudes that are similar to the other models. The only time interval that has a significant hazard baseline function at 10% is the one from July 1996 to December 1996, and the conditional probability of failure is smaller than in any other period. An interesting result in Model C is that the variable " import/ international reserve ratio" increases the probability of failure in the banking system.

We estimate other parameterizations of the baseline hazard function for models A, C and D: Weibull, Log-logistic, Lognormal, Gompertz, and generalized Gamma¹⁵ (Appendix A.2). We also estimate the non-parametric Cox model using only banklevel variables. Based on the AIC and BIC, the exponential model is in general better specified, and whenever one of these criteria pointed to some other specification, the predictive capability of the exponential model was always better.

Figure 2 shows the estimated survival functions to model A (exponential model, including micro and macro variables) for the banks that fail and for the ones that do not fail. Because we have a survival function for each bank, we calculate the average probability of survival for each time period $(1, 2, \ldots, T)$, for each group (failed and non-failed). We notice that the survival dynamics is substantially different for the

¹⁵There is no equivalence to model B (piecewise-exponential) for these other parameterizations.

two groups. The survival function for the failed banks is always below the one for the non-failed banks, but they separate after a 30-month period, in December 1996 after Proer implementation.



Figure 2: Estimated In-Sample Survival Functions

In order to verify the model's goodness of fit, we plot the predicted Cox-snell residuals (Newmann, 1997). If the model represents a good fit, the residuals should have approximately an exponential distribution with unit parameter. We treat the residuals as a time variable and plot them against the accumulated empirical hazard distribution. A good fit would result in a straight line with slope equal to one. Figure 3 presents these plots for models A and D.¹⁶ All exponential and PCE models seem to be well adjusted. The right tail of the distribution presents higher dispersion because of the reduction in the sample size caused by failure and censoring.



¹⁶The plots of residuals for the other models are available from the authors upon request.

Figure 3: Cox-snell residuals

Figure 4 shows Martingale residuals versus A201 covariate (ratio of non - performing loans to total loans) to Model A. Plots of those residuals are useful in assessing whether the functional form of the covariate is adequate. As we can see, as the smooth curve is linear around zero, no transformation is necessary. The same is true to the other covariates and models.



Figure 4: Martingale residuals - Model A

6.2 Forecasting Bank Failure

6.2.1 In-Sample Forecasting

Most studies about bank fragility make in-sample forecasts for certain time intervals. Actually, since most studies assume fixed covariates, the forecasted survival function depends directly on the chosen hazard baseline function, which varies in time. The forecast for intervals is used when one does not have new information available in each period to distinguish the individuals (banks), but this does not happen when panel data is used.

Using the estimated models to classify banks as failed and non-failed, for each time horizon, in and out of sample, is not trivial. The estimated survival probabilities should be compared to some critical (cutoff) value. Typically, the proportion of banks that fail and do not fail in the sample is used to determine this cutoff.¹⁷ This

 $^{^{17}\}mathrm{See}$ for example Dabós and Escudero (2000) and Lane et al. (1986).

approach has some drawbacks when the covariates vary in time or even when the data is censored (banks leave or enter the sample during the period), depending on how the failures are distributed in time. If we do not adjust the sample for the censored banks, we would overestimate the proportion of failures. In this sense, the model would say less often that a bank fails, diminishing the probability of type II error and increasing the type I error.

Because of the use of panel data, we are able to forecast the probability of bank failure in each month. We calculate the proportion of failures and non-failures, month by month, correcting for censoring which also includes the banks that are born in the period. The proportion of banks that fail in the first period is given by the ratio of number of failures in the first period to the sum of failures and censored banks after the first period. The same reasoning is applied to the subsequent periods. However, these proportions are the ones used for the Kaplan-Meier estimates. Therefore, our comparisons are between the estimated survival function and the proportion of failures corrected month by month.

The null "hypothesis" in our analysis is a bank failure, and we are interested in type I error (the model does not predict a bank failure when it actually happens), but also in type II error (the model predicts a failure when it does not happen). From the regulator point of view, type I errors should be given a higher weight in his objective function because it involves higher costs to resolve. However, type II errors also imply in costs, given that the regulator has scarce resources and has to prioritize a group of banks to supervise.

Characterizing a prediction as type II error depends on the period and time horizon of interest. Some type II errors represent banks that fail at sometime in the near future, and therefore could be considered a model success. The in-sample forecasts for all model specifications are presented in Table VI.

The first cell in Table VI indicates that model A forecasts correctly the month in which the failure occurs 73% of the time, during the considered period. Considering the banks that actually fail during the period, but taking only the ones for which the

relationship between the events forecasted as a failure and the events not forecasted as failure is above the percentage cutoff for type II error, the type I error decreases to 3%. In this case, the model forecasts correctly 97% of the time in a 53-month time horizon.¹⁸

Forecast Errors	Model A	Model B	Model C	Model D
Type I Error	0.27	0.29	0.30	0.38
Type I Error in 53 Months	0.03	0.03	0.03	0.03
Type II Error	0.30	0.31	0.32	0.26
Type II Error for d=1	0.54	0.56	0.58	0.44

Table VI: In-Sample Forecasts from 07/1994 to 12/1998

Note: d=1 indicate the banks that fail at sometime of the considered period, but not in the specific month.

Since type II errors could be of interest (prediction of a failure when it dos not happen), this percentage, for banks that fail in the period, but not in a specific month, is higher (0.54 instead of 0.30) than for the banks that do not fail, i.e., the model predicts correctly that the banks which eventually fail had problems before hand. This makes it possible for the regulator to act ex-ante, reducing the costs of resolving a bank failure, including the costs of rediscount lending from the Central Bank.

Model D (only includes microeconomic variables) is the one with the worst forecasting power, given that its type I error is greater than any other model's. On the other hand, model D has the lowest probability of type II error, but it cannot distinguish very well the failed from the non-failed banks, as the difference between the two kinds of type II error is the smallest.

6.2.2 Out-of-Sample Forecasting

It would be ideal to use the estimated models for July 1994 to December 1998 to predict the subsequent years. However, the INDCON system was interrupted after

¹⁸For the banks that do not fail, using the same cutoff value and the same methodology, the percentage of type II error (around 0.30) is similar to the one obtained for the considered period.

1999, making it impossible to perform forecasts after this year. We decide to reestimate the models using the July 1994 to December 1997 data and simulate the forecast for January to December of 1998.

The first issue to be discussed here is the cutoff value for the survival probability. Some authors (see Whalen, 1991; and Dabós and Escudero, 2000) adopt the same cutoff used for the in-sample forecast, or they calculate another cutoff taking into account only the banks that fail during the time period outside the sample. We make use of the Kaplan-Meier (KM) estimates for the period until 1997 and we construct new probabilities taking into account only the failures that happen in 1998.¹⁹

Forecast Errors	Model A	Model B	Model C	Model D
Type I Error in the month	0.33	0.40	0.40	0.33
Type I Error in 12 Months	0.17	0.17	0.25	0.33
Type II Error	0.30	0.34	0.28	0.32
Type II Error for d=1	0.40	0.40	0.37	0.43

Table VII: Out-of-Sample Forecasts from 01/1998 to 12/1998

Note: d=1 indicate the banks that fail at sometime of the considered period, but not in the specific month.

Table VII presents the out-of-sample forecasts. Models A and D forecast correctly the month of a failure in 67% of the cases. Models A and B make the best forecast if we consider a one year period, 83% of the cases. Note that a one-year period forecast represents an upper bound for the out-of-sample forecast error. The predictions for a two-year period should be better and it would improve for longer periods of time.

6.3 Comparison of Bank Failure Models

In order to compare our results with previous duration models for bank failure in Brazil, we estimate a model including only microeconomic variables and cross-section data from July 1994. Table VIII shows the results. There are only four significant

¹⁹The results are only slightly different than if we use the KM for the period until 1998 or if we consider the survival function for period until 1997.

variables in this model, and all except "evolution of typical assets operations" are also significant in our panel data models with micro, macro and contagion variables. The forecast accuracy of this model is a lot worse that of our models. For the insample forecast, the type I error for a 53-month period is 0.21, predicting a failure in 79% of the cases, while it is 0.03 for our model A, predicting a failure in 97% of the cases. The type II error is 0.42 and it is 0.30 for our model A.

	11		
	Hazard	s.e.	
Variables	Ratio	Robust	P> z
Recovery Adm. Expenses	0.9922	0.0075	0.300
Atypical assets/ total assets***	1.0467	0.0130	0.000
Operational Margin	1.0102	0.0149	0.491
Leverage ratio*	1.0001	0.0001	0.068
Non-performing/ total loans	1.0191	0.0186	0.299
Evol. typical assets operations*	1.0002	0.0001	0.091
Loan reserve coverage	0.9978	0.0247	0.928
Adm. Cost/ average assets	1.1610	0.1166	0.137
Other liabilities/ tot liabilities***	0.9694	0.0114	0.008
Return on adjusted tot assets	0.8524	0.0966	0.159
Final real assets	1.0000	1.24E-10	0.582

Table VIII: Cross-Sectional Model- Covariates from July 1994

Note: ***, ** and * indicates significance at 1%, 5% and 10% respectively.

7 Applications

7.1 Mean and Median Survival Time

The duration models developed in this paper can be applied to obtain the estimated (in-sample) time of bank failures. Using each conditional model we estimate the mean and median survival times of individual banks for each month of the 1994-1998 period. The median survival time is defined as the time, t, for which $\hat{S}(t) = 0.5$ while the mean survival time is defined as:

$$Tmean = \int_{0}^{\infty} \hat{S}(t)dt,$$

where $\hat{S}(t)$ is the estimated (in-sample) survival function.

Figure 5 reports the mean and median survival times according to Model A by bank category: 0 represents the banks that survived through December 1998 and 1 the banks that failed in some period until December 1998. There is a concentration of short survival times for failed banks. As one would expect, the surviving banks exhibit long survival time very often. A striking result is that the failed banks that display long survival time do so only in specific months and, surprisingly, just before the bankruptcy event²⁰.



Figure 5: Mean and Median Survival Times - Model A - by Category

Besides making bank-level surveillance of the banking system possible, the expected time to failure allows the regulator to oversee the survival time path and to detect outliers so that it becomes possible to take remedial action. In addition the estimated survival times are comparable among all banks, making any segmentation useless.

7.2 Financial Fragility Index

The methodology proposed in this paper can be extended to provide an indicator of system-wide financial fragility. We build a financial fragility index (IFF) based on

 $^{^{20}}$ A possible explanation is that when a bank is near to fail it tends to hidden information of its account system.

the estimated (in-sample) conditional probability of failure of each individual bank for each month. The index is given by:

$$IFF_t = \sum_i p_{it} \alpha_{it}$$

where p_{it} is the conditional probability of failure, "t" indexes month, "i" indexes bank and α_{it} is the ratio of each bank's assets to the total assets of the banking system.

Figure 6 displays the IFF based on Models A (micro and macro covariates) and D (only micro covariates) estimations from July 1994 to December 1998. The IFF follows a similar path for both models, except for the time interval between 35 and 40 months when it is declining for Model A and increasing for Model D. Model A presents a local maximum around 35 months, time of Banco Bamerindus failure. Both models indicate a high degree of banking fragility soon before the adoption of the floating exchange rate regime (around the 50^{th} month).



Figure 6: Financial Fragility Index

8 Contagion

In this section, we approach the question of the likelihood that banks failed during the Brazilian banking crisis because of contagion in the banking system. Our case study is the Proer. There was a debate during the institution of Proer about whether the Central Bank should grant subsidized loans to the banks after the December 1995 crisis. The prevalent argument was that the increased systemic risk would make *ex-ante* solvent banks fail without that liquidity support.

In our case it is difficult to analyze the effectiveness of Proer in avoiding contagion, since a counterfactual (bank failure without Proer) does not exist. Because it is a universal program (all banks can benefit from it) there is no control group to perform an evaluation of the program. However, if we assume that contagion would happen, we can verify if Proer was efficient in stopping it²¹. Also, we would like to look for evidence that contagion would actually take place without Proer.

We divide our sample of banks in three groups: banks that failed until November 1995 (before Proer) – group 1, banks that fail during Proer – group 2, and banks that survived through December 1998 – group 0. We then compare *ex-ante* attributes of the groups, testing for differences across groups in means and medians of: i) the estimated probability of failure of banks; ii) the monthly real demand deposits growth.

Testing if group 2 banks are on average as strong as survivors (in terms of probability of failure) amounts to say that solvent banks could have failed during Proer. In addition, if group 2 banks are at least as strong as group 1 banks, we would say that there is evidence in favor of the null hypothesis that bank failures were a continuation of the same process and that Proer may not have reduced systemic risk.

Table IX indicates that banks that failed during Proer and survivor banks have different (mean and median) monthly real demand deposits growth, but banks that failed before Proer and survivors have not. This is evidence that before Proer depositors were not able to distinguish between solvent and insolvent banks and that Proer was effective in avoiding bank runs.

 $^{^{21}}$ Here, we refer to contagion by asymmetric information as in De Bandt and Hartmann (2000).

Table IX: Tests for Differences in Mean and Median:

Group	Mean ^{1/}	Median
0	0,0328	0,0063
1	0,0086	-0,0059
2	0,0123	-0,0048
Statistics t and χ^2		
0 e 1	0,5425	0,2215
0 e 2	1,8693*	13,8876***
1 e 2	-0,0803	0,0096

Monthly Real Demand Deposits Growth

 T Bartlett's test was applied to confirm if the two-sample data have equal variances. If unequal the test was carried on using the Satterthwaite approximation.

The tests for differences in the conditional probability of failure are presented in Table X. The mean and median probabilities of failure are significantly different for all three groups of banks. The survivors have the lowest probability of failure when compared to any of the other groups. The banks that fail during the Proer seem to be stronger than the ones that fail before, which could be an indication that solvent banks break during the Proer.

Group	Mean	Median	
0	0,0054	0,0036	
1	0,0149	0,0063	
2	0,0093	0,0042	
Statistics t e χ^2			
0 e 1	-3,7069***	64,1114***	
0 e 2	-5,0346***	29,2685***	
1 e 2	2,2430 **	39,9466***	

 Table X: Tests for Differences in Mean and Median

 Monthly Conditional Probability of Failure – Model A

¹⁷ Bartlett's test was applied to confirm if the two-sample data have equal variances. If unequal the test was carried on using the Satterthwaite approximation.

Based on the demand deposits growth rate it seems that depositors could distinguish between solvent and insolvent banks during the Proer, and this may have contributed to avoid bank runs and contagion. However, as we reject the hypothesis of equality in the conditional probability of failure among groups, we need some caution in interpreting the results in terms of contagion.

9 Conclusion

Based on empirical duration models we estimate conditional probabilities of failure for all Brazilian banks, except foreign-owned ones, that existed in July 1994 or were born between this month and December 1998. We also aggregate all conditional probabilities of failure in a single index (IFF) – Financial Fragility Index – that gives a general measure of crisis risk.

Our analysis finds that foreign banks have a very particular survival behavior, and therefore should be studied separately. The bank-level covariates in our conditional models were significantly related to the probability of failure, however, none of the liquidity indicators were relevant.

This study has the following main contributions. First, it adds contagion and macroeconomic variables to the traditional models based upon accounting data. These macro variables are in fact significant in some models and contribute to increase the predictive power of these duration models.

Second, it uses panel data. The introduction of time-varying variables increases enormously the predictive power of the models relatively to similar specified crosssection models. Some of our models are able to predict the exact month a failure occurs 73% of the time (in-sample forecast). If we are only concerned about failure and not the exact month, the model is correct 97% of the time.

Third, the study discusses the effectiveness of the program Proer in reducing bank runs and contagion. Even though it is not possible to formally test the hypothesis that Proer avoided contagion, we find indications that Proer was efficient in separating solvent from insolvent banks *ex-ante*, minimizing the failure of sound banks.

Fourth, the problem of left censoring in duration models data was addressed. The non-consideration of the actual date of birth of banks can bias and decrease the forecasting accuracy of the models.

One final contribution is the demonstration that bank-level indicators are not necessarily poor predictors of bank failure. Our innovation is the use of a broader dataset.

References

- Almeida, Mansueto F. & J. R. Mendonça de Barros. "An Analysis of the Financial Adjustment in Brazil", Secretaria de Política Econômica, April 1997.
- [2] Arena, Marco A. "Bank Fundamentals, Bank Failures and Market Discipline: An Empirical Analysis for Emerging Market During the Nineties". Paper presented at LACEA, 2004.
- [3] Banco Central do Brasil. "Manual INDCON Sistema de Indicadores Econômico-Financeiros de Instituições sob Acompanhamento e Controle do Banco Central", Decad, 1995.
- [4] Banco Central do Brasil. "O Sistema INDCON Indicadores Econômico-Financeiros de Instituições sob Acompanhamento e Controle do Banco Central do Brasil", 1994.
- [5] Bercoff, José J., Julian di Giovanni & Franque Grimard. "Argentinean Banks, Credit Growth and the Tequila Crises: A Duration Analysis". Mimeo, July 2002.
- [6] Bernanke, B., & M. Gertler. "Financial Fragility and Economic Performance". Quarterly Journal of Economics 105(1), 1990, p. 87-114.
- [7] Calomiris, Charles W. & Andrew Powell. "Can Emerging Market Bank Regulators Establish Credible Discipline? The Case of Argentina, 1992-1999". NBER Working Paper n^o 7715, May 2000.

- [8] Calomiris, Charles & Joseph R. Mason. "Contagion and Bank Failures During the Great Depression: The June 1932 Chicago Banking Panic," American Economic Review 87(5), 863-883, December 1997.
- [9] Cole, Rebel A., & Jeffery W. Gunther. "Separating the Likelihood and Timing of Bank Failure", Journal of Banking and Finance 19, 1995, 1073-1089.
- [10] Cox, D. R. "Regression Models and Life-Tables", Journal of the Royal Statistical Society, Series B 34, 187-220, May/Aug. 1972.
- [11] Cysne, Rubens Penha e Sérgio G. Silveira da Costa. "Effects of the Real Plan on the Brazilian Banking System, EPGE Ensaios Econômicos, nº 304, Maio de 1997.
- [12] D'Amato, Laura, Elena Grubisic & Andrew Powell. "Contagion, Bank Fundamentals or Macroeconomic Shock? An Empirical Analysis of the Argentine 1995 Banking Problems". Working Paper n^o 2, Central Bank of Argentina, July 1997.
- [13] Dabós, Marcelo & Walter Sosa Escudero. "Explaining and Predicting Bank Failure in Argentina Using Duration Models", Mimeo, April 2000.
- [14] De Bandt, O. & P. Hartmann. "Systemic Risk: A Survey", Discussion Paper nº 2634, Centre for Economic Policy Research, 2000.
- [15] Demirgüç-Kunt, Asli & E. Detragiache. "Monitoring Banking Sector Fragility: A Multivariate Logit Approach with an Application to the 1996-97 Banking Crises". Mimeo, January 1999.
- [16] Demirgüç-Kunt, A. & E. Detragiache. "The Determinants of Banking Crises in Developing and Developed Countries", IMF Working Paper vol. 45, n^o 1, 1998.
- [17] Demirgüç-Kunt, Asli. "Principal-Agent Problems in Commercial-Bank Failure Decision", Working Paper n^o 9106, Federal Reserve Bank of Cleveland, 1991.

- [18] DeYoung, Robert. "For How Long Are Newly Chartered Banks Financially Fragile?" WP 2000-09, Federal Reserve Bank of Chicago, 2000.
- [19] Diamond, D. & P. Dybvig. "Bank Runs, Deposit Insurance, and Liquidity", Journal of Political Economy, no 91 (3), 1983.
- [20] González-Hermosillo, Brenda. "Determinants of Ex-Ante Banking System Distress: A Macro-Micro Empirical Exploration of Some Recent Episodes", IMF Working Paper, March 1999.
- [21] González-Hermosillo, Brenda, Ceyla Pazarbasioglu, &Robert Billings. "Determinants of Banking System Fragility: A Case Study of Mexico," IMF Staff Papers, 1997.
- [22] Kalbfleisch, J. D. & R. L. Prentice. The Statistical Analysis of Failure Time Data. New York: John Wiley & Sons, 1980.
- [23] Kiefer, Nicholas M. "Economic Duration Data and Hazard Functions", Journal of Economic Literature, vol. 26, n° 2, June 1988, 646-679.
- [24] Klein, J. P. & M. L. Moeschberger. Survival Analysis: Techniques for Censored and Truncated Data. New York, Springer-Verlag, 1997.
- [25] Neumann, George R. "Search Models and Duration Data", in M. Hashem Pesaran and Peter Schmidt, eds., Handbook of Applied Econometrics: Microeconomics, Vol. II, chapter 4, Oxford, Blackwell Publishers, 1997, 300-351
- [26] Janot, Marcio Magalhães, "Modelos de Previsão de Insolvência Bancária no Brasil", Working Paper nº 13, Banco Central do Brasil, Março 2001.
- [27] Lancaster, Tony. The Econometric Analysis of Transition Data, Econometric Society Monographs n^o 17, Cambridge University Press, 1990.
- [28] Lane, William R., Stephen W. Looney & James W. Wansley, "An Examination of Misclassifications with Bank Failure Prediction Models", Journal of Economics and Business, 1989, 327-336.

- [29] Lane, William R., Stephen W. Looney & James W. Wansley, "An Application of the Cox Proportional Hazards Model to Bank Failure", Journal of Banking and Finance 10, 1986, pp. 511-531.
- [30] Lopes, Francisco L. "Notes on the Brazilian Crisis of 1997-99". Brazilian Journal of Political Economy, vol. 23, n^o 3 (91), 2003.
- [31] Rocha, Fabiana F. "Previsão de falência Bancária: Um Modelo de Risco Proporcional", Pesquisa e Planejamento Econômico, vol. 29, nº 1, 1999, 137-152.
- [32] StataCorp. Stata Statistical Software: Release 7.0. College Station, TX, 2001.
- [33] Rochet, Jean-Charles. "Bank Runs and Financial Crises". Mimeo, 2002.
- [34] Rojas-Suarez, Liliana. "Rating Banks in Emerging Markets: What Credit Rating Agencies Should Learn From Financial Indicators", Institute for International Economics Working Paper, June 2001.
- [35] Whalen, G. "A Proportional Hazards Model of Bank Failure: An Examination of Its Usefulness as an Early Warning Tool", Federal Reserve Bank of Cleveland Economic Review, Vol. 27, No. 1, 1991.
- [36] Wheelock, David C. & Paul W. Wilson. "Why Do Banks Disappear: The Determinants of U. S. Bank Failures and Acquisitions", Review of Economics and Statistics 82(1), February 2000, 127-38.
- [37] Wheelock, David C. & Paul W. Wilson. "Explaining Banking Failures; Deposit Insurance, Regulation, and Efficiency", The Review of Economics and Statistics, vol. 77, n^o 4, 1995, 689-700.
- [38] Wooldridge, Jeffrey M. Econometric Analysis of Cross-Section and Panel Data. The MIT Press, Cambridge, 2002.

A Appendix

A.1 Description of the Variables

Indicator	What the Variables Measure	Survival Time: Expected Sign
A101 - ratio of liquid assets to typical	Liquidity risk	_/+. High ratio may indicate that the
assets		bank has difficulties to participate in
		the interbank market. On the other
		hand, it may measure bank's ability
		to deal with deposit withdrawals.
A102 - ratio of assets to adjusted total	Fraud and market risks	+
assets		
A103 - ratio of atypical assets to total	Fraud Risk	-
assets		
A106 - ratio of real state loans to total	Market risk	Commercial real state loans tend
loans		to be risky because they typically
		have long maturation periods.
A108 - ratio of foreign exchange	Market risk	Negative if the ratio indicates high
operations to typical asset operations		concentration of foreign exchange
		operations.
A201 - ratio of non-performing loans to	Credit risk	-
total loans		
A202 - financial assets reserve coverage	Credit risk	+
A203 - loan reserve coverage	Credit risk	_/+ . Growing trend may indicate
		deterioration of the bank's loans
		quality. High level of reserves may
		represent a cushion to absorb shocks.
A301 - ratio of the total funding to	Mismatching between assets	-
typical assets	and liabilities	
A302 - ratio of the non-interest-bearing	Liquidity risk	-
deposits to typical assets		
A303 - ratio of interest-bearing deposits	Liquidity risk	_/+. It depends on the deposit being
to total asset operations		more or less volatile during crises.
C102 – ratio of loans funded by the	Credit risk	_/+. It depends on the business cycle.
domestic market to liabilities		
C104 - ratio of loans funded by the	It should be composed with it	_ /+. It depends on the business
foreign market to liabilities	above.	cycle.

Bank-Level Indicators

Indicator	What the Variables Measure	Survival Time: Expected Sign
C107 - ratio of other liabilities to total	Credit risk	_
liabilities.		
C204 - leverage ratio	Credit risk	_
E202 - evolution of typical assets	To verify lending booms	_/+ . Negative if it indicates lending
operations		booms.
L104 - solvency	Solvency	+
R102 - monthly average return on equity	Solvency	+
in the semester		
R201 - net margin	Efficiency	+
R205 - recovery of the administrative	Efficiency	+
expenses through service's income		
R206 - return on adjusted total assets	Efficiency of the investment	_/+. High yields may indicate that the
	or increase of credit risk if	bank is taking risky loans. Low yield
	loans becomes higher	may indicate that that risk is not
		priced properly.
R301 - operational margin (difference	Spread	-
between loan yield and deposit interest		
rate) *		
R305 - operational margin (monthly	Spread	-
average in the semester)		
R308 - monthly average yield on	Profitability	_ /+ . It depends on the state of the
operational assets in the semester		banking cycle because risky projects
		can be very profitable at first.
R401 - interest income earned by term	Credit risk	-
depositors		
R403 - ratio of administrative cost to	Efficiency	_/+. It depends on the bank's profile
adjusted total assets		(if retailer or wholesaler)
R405 – total funding cost	Credit risk	-
R406 - ratio of salaries and employee	Efficiency	-
benefits to managerial expenses		
R409 - ratio of administrative cost to	Efficiency.	-
average assets		

Bank-Level Indicators (cont.)

^{*} In developed countries, the sign would usually be positive because it would mean that the bank is efficient. In emerging countries, high spreads may mean that the bank is "gambling for the resurrection" as Rojas-Suárez (2001) displays: high spreads show that the bank is making risky loans.

Macroeconomic Covariates

Variables	Description	Survival Time: Expected Sign
IPCA	Consumer Price Index	-
IPI	Industrial Production Indicator $(2002 = 100)$.	_/+. Negative if it lags lending
		booms.
Selic	Selic rate accumulated in the month per year	_ /+. Negative if it is related to
	and deflated by IPCA	potential interest rate shock.
Embi	Average spread over US Treasury of Brazilian	-
	sovereign bonds calculated by J.P. Morgan	
Mres	Ratio of total imports to international reserves	_/+
	(liquidity concept)	
M1res and M2res	Ratio of means of payment - M1 (average in	_
	the working days of the month) and M2 (stock	
	at the end of month) to international reserves	
	(liquidity concept), respectively	

Contagion Covariates

Indicator	What the Variables Measure	Survival Time: Expected Sign		
Crisco **	Ratio of risk 2 loans (level H) to total loans	Potential bank herding behavior		
		or deposit runs.		
Crgdp	Ratio of total loans (in USD) to monthly GDP	_/+ . If the ratio is very high, it		
	(in USD).	may indicate regulatory		
		forbearance.		
Varcre	Percentual change of loan per month	It may indicate lending booms.		

Specific Covariates

Indicator	What the Variables Measure	Survival Time: Expected Sign
ativoreal (real assets)	Total assets deflated by IPCA.	+. Economies of scale and portfolio
		diversification.
Vardvlag3 and Vardvmean	Monthly changing of demand deposits.	+ . Higher demand deposit means
		more creditworthiness.

** Risk 2 loans are loans with past due more than 60 days and without enough collateral and for all loans with past due 180 days or more.

A.2 Results

A.2.1 Equality of Survival Functions

Log-rank				
Туре	Observed	Expected		
	events	events		
	9	6.62		
	2 44	33.62		
	3 1	11		
	4 5	7.76		
Total	59	59		

chi2(3) = 14,25

Pr>chi2 = 0,0026

Log-rank

Туре	Observed events	Expected events
1	9	7.99
2	44	40.64
4	5	9.37
Total	58	58

chi2(2) = 2,47

Pr>chi2 = 0,2913

A.2.2 Descriptive Statistics

Desciptive statistics and test for equality of mean

Variable	Mean		Median		Standard-erro	r (mean)
	d=0	d=1	d=0	d=1	d=0	d=1
r205	29.12	35.81	8.81	6.84	1.044	7.790
a103*	6.45	7.50	3.01	4.50	0.108	0.262
r305***	-114.06	0.98	0.57	0.81	10.785	0.195
c204***	853.62	1282.46	520.37	551.80	14.006	60.791
a201	13.10	13.27	6.00	5.69	0.199	0.502
e202***	5.20	35.84	6.18	0.59	9.471	7.754
a203***	9.00	6.94	3.27	2.16	0.173	0.347
r409***	0.84	1.24	0.59	0.84	0.008	0.040
c107**	21.44	73.21	7.81	8.68	2.513	19.820
r206***	-0.19	-0.42	0.11	0.03	0.036	0.096
ativorealfim***	2.10E+08	1.40E+08	5.70E+06	2.20E+06	1.58E+07	1.78E+07
a101	20.86	-6.78	40.14	36.83	5.042	18.816
a102***	89.17	85.58	93.21	90.45	0.135	0.349
a106***	5.43	3.24	0.00	0.00	0.162	0.220
a108	10.90	7.34	2.06	0.23	0.174	0.340
a202	3.38	2.67	0.20	0.22	0.114	0.179
a301***	79.56	85.61	85.32	89.53	0.474	0.750
a302***	3.88	6.83	0.80	0.86	0.089	0.388
a303	141.23	224.08	77.47	87.49	8.208	29.422
c102***	62.25	70.51	68.79	76.04	0.283	0.666
c104***	17.28	9.83	5.32	0.18	0.245	0.416
1104***	255.55	189.73	120.61	120.74	7.148	12.438
r102	-0.36	-0.35	0.92	0.58	0.295	0.166
r201***	-125.33	-488.24	5.77	1.42	60.971	153.448
r301***	-153.27	-45.30	0.61	0.83	12.456	17.513
r308	7.94	3.78	2.78	2.99	1.496	0.200
r401***	150.83	4.16	0.00	-0.32	12.242	1.447
r403***	0.98	1.30	0.60	0.80	0.020	0.046
r405***	27.94	2.62	-1.06	-1.22	5.089	1.448
r406	6.13	-16.35	51.87	50.86	12.007	40.914

Obs₁: d=0 for banks that did not fail and d=1 otherwise.

 $Obs_{2}:\ ^{*} \ \text{statistically significant to } 0,10;\ ^{**} \ \text{statistically significant to } 0,05;\ ^{***} \ \text{statistically significant to } 0,01.$

A.2.3 Factorial Analysis



Proportion of total variance

Factor	Proportion	Accumulated
1	0.117	0.117
2	0.091	0.208
3	0.076	0.284
4	0.063	0.347
5	0.050	0.397
6	0.048	0.445
7	0.041	0.486
8	0.039	0.524
9	0.037	0.562
10	0.037	0.598
11	0.035	0.633

A.2.4 Models B and D

	Ν	Model B			Model C	
	Hazard	s.e.		Hazard	s.e.	
Variables	ratio.	Robust	P> z	ratio	Robust	P> z
e2	0.61123	0.32539	0.355			
e3	0.88446	0.43497	0.803			
e4	0.51569	0.28879	0.237			
e5*	0.22861	0.17885	0.059			
e6	0.83231	0.41070	0.710			
e7	0.81028	0.41939	0.684			
e8	1.14667	0.57368	0.784			
e9	0.20208	0.20743	0.119			
varcre2**				1.02919	0.01159	0.011
mres2***				2.9E+8	2.0E+09	0.006
Recovery Adm. Expenses**	0.98906	0.00535	0.042	0.99021	0.00452	0.031
Atypical assets/ total assets ***	1.03252	0.00782	0.000	1.03423	0.00766	0.000
Operational Margin***	1.00045	0.0001	0.000	1.00045	0.00010	0.000
Leverage ratio***	1.0002	0.00005	0.000	1.0002	0.00005	0.000
Non-performing/ total loans ***	1.03274	0.00531	0.000	1.03368	0.00493	0.000
Evol. typical assets operat. * ^{1/}	0.99995	0.00005	0.336	0.99996	0.00004	0.324
Loan reserve coverage***	0.96639	0.00921	0.000	0.96586	0.00869	0.000
Adm. Cost/ average assets	1.11309	0.09027	0.186	1.09195	0.09507	0.312
Other liabilities/ liabilities.***	1.00032	0.00005	0.000	1.00032	0.00005	0.000
Return on adjusted total assets	0.97571	0.01905	0.208	0.97804	0.01909	0.255
Real assets	1.00000	0.00000	0.182	1.00000	4.4E-10	0.182

Summary of Results for Models B and D

Note: ***, ** and * indicates significance at 1%, 5% and 10% respectively.

A.2.5 Model and Distribution Selection

Criteria AIC e BIC to model selection

Model	df	AIC	BIC
A	14	300.46	400.23
В	20	307.55	450.08
С	14	299.20	398.96
D	12	302.27	387.78

Criteria AIC e BIC to distribution selection

Distribution	Model A		Model C		Model D	
Criteria	AIC	BIC	AIC	BIC	AIC	BIC
Exponential	300.47	400.23	299.20	398.96	302.27	387.78
Weibull	301.62	408.51	297.90	404.79	304.26	396.90
Log-logistic	302.35	409.24	295.76	402.65	304.46	397.10
Lognormal	300.11	407.00	295.13	402.02	304.10	396.74
Gompertz	299.13	406.03	300.68	407.57	303.99	396.63
Gammma Gen.	298.90	412.92	295.73	409.75	302.86	402.63

Banco Central do Brasil

Trabalhos para Discussão

Os Trabalhos para Discussão podem ser acessados na internet, no formato PDF, no endereço: http://www.bc.gov.br

Working Paper Series

Working Papers in PDF format can be downloaded from: http://www.bc.gov.br

1	Implementing Inflation Targeting in Brazil Joel Bogdanski, Alexandre Antonio Tombini and Sérgio Ribeiro da Costa Werlang	Jul/2000
2	Política Monetária e Supervisão do Sistema Financeiro Nacional no Banco Central do Brasil Eduardo Lundberg	Jul/2000
	Monetary Policy and Banking Supervision Functions on the Central Bank Eduardo Lundberg	Jul/2000
3	Private Sector Participation: a Theoretical Justification of the Brazilian Position <i>Sérgio Ribeiro da Costa Werlang</i>	Jul/2000
4	An Information Theory Approach to the Aggregation of Log-Linear Models <i>Pedro H. Albuquerque</i>	Jul/2000
5	The Pass-Through from Depreciation to Inflation: a Panel Study Ilan Goldfajn and Sérgio Ribeiro da Costa Werlang	Jul/2000
6	Optimal Interest Rate Rules in Inflation Targeting Frameworks José Alvaro Rodrigues Neto, Fabio Araújo and Marta Baltar J. Moreira	Jul/2000
7	Leading Indicators of Inflation for Brazil <i>Marcelle Chauvet</i>	Sep/2000
8	The Correlation Matrix of the Brazilian Central Bank's Standard Model for Interest Rate Market Risk José Alvaro Rodrigues Neto	Sep/2000
9	Estimating Exchange Market Pressure and Intervention Activity Emanuel-Werner Kohlscheen	Nov/2000
10	Análise do Financiamento Externo a uma Pequena Economia Aplicação da Teoria do Prêmio Monetário ao Caso Brasileiro: 1991–1998 Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Mar/2001
11	A Note on the Efficient Estimation of Inflation in Brazil <i>Michael F. Bryan and Stephen G. Cecchetti</i>	Mar/2001
12	A Test of Competition in Brazilian Banking Márcio I. Nakane	Mar/2001

13	Modelos de Previsão de Insolvência Bancária no Brasil Marcio Magalhães Janot	Mar/2001
14	Evaluating Core Inflation Measures for Brazil Francisco Marcos Rodrigues Figueiredo	Mar/2001
15	Is It Worth Tracking Dollar/Real Implied Volatility? Sandro Canesso de Andrade and Benjamin Miranda Tabak	Mar/2001
16	Avaliação das Projeções do Modelo Estrutural do Banco Central do Brasil para a Taxa de Variação do IPCA Sergio Afonso Lago Alves	Mar/2001
	Evaluation of the Central Bank of Brazil Structural Model's Inflation Forecasts in an Inflation Targeting Framework <i>Sergio Afonso Lago Alves</i>	Jul/2001
17	Estimando o Produto Potencial Brasileiro: uma Abordagem de Função de Produção <i>Tito Nícias Teixeira da Silva Filho</i>	Abr/2001
	Estimating Brazilian Potential Output: a Production Function Approach <i>Tito Nícias Teixeira da Silva Filho</i>	Aug/2002
18	A Simple Model for Inflation Targeting in Brazil Paulo Springer de Freitas and Marcelo Kfoury Muinhos	Apr/2001
19	Uncovered Interest Parity with Fundamentals: a Brazilian Exchange Rate Forecast Model Marcelo Kfoury Muinhos, Paulo Springer de Freitas and Fabio Araújo	May/2001
20	Credit Channel without the LM Curve Victorio Y. T. Chu and Márcio I. Nakane	May/2001
21	Os Impactos Econômicos da CPMF: Teoria e Evidência <i>Pedro H. Albuquerque</i>	Jun/2001
22	Decentralized Portfolio Management Paulo Coutinho and Benjamin Miranda Tabak	Jun/2001
23	Os Efeitos da CPMF sobre a Intermediação Financeira Sérgio Mikio Koyama e Márcio I. Nakane	Jul/2001
24	Inflation Targeting in Brazil: Shocks, Backward-Looking Prices, and IMF Conditionality Joel Bogdanski, Paulo Springer de Freitas, Ilan Goldfajn and Alexandre Antonio Tombini	Aug/2001
25	Inflation Targeting in Brazil: Reviewing Two Years of Monetary Policy 1999/00 Pedro Fachada	Aug/2001
26	Inflation Targeting in an Open Financially Integrated Emerging Economy: the Case of Brazil Marcelo Kfoury Muinhos	Aug/2001
27	Complementaridade e Fungibilidade dos Fluxos de Capitais Internacionais Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Set/2001

28	Regras Monetárias e Dinâmica Macroeconômica no Brasil: uma Abordagem de Expectativas Racionais <i>Marco Antonio Bonomo e Ricardo D. Brito</i>	Nov/2001
29	Using a Money Demand Model to Evaluate Monetary Policies in Brazil <i>Pedro H. Albuquerque and Solange Gouvêa</i>	Nov/2001
30	Testing the Expectations Hypothesis in the Brazilian Term Structure of Interest Rates <i>Benjamin Miranda Tabak and Sandro Canesso de Andrade</i>	Nov/2001
31	Algumas Considerações sobre a Sazonalidade no IPCA Francisco Marcos R. Figueiredo e Roberta Blass Staub	Nov/2001
32	Crises Cambiais e Ataques Especulativos no Brasil <i>Mauro Costa Miranda</i>	Nov/2001
33	Monetary Policy and Inflation in Brazil (1975-2000): a VAR Estimation André Minella	Nov/2001
34	Constrained Discretion and Collective Action Problems: Reflections on the Resolution of International Financial Crises Arminio Fraga and Daniel Luiz Gleizer	Nov/2001
35	Uma Definição Operacional de Estabilidade de Preços <i>Tito Nícias Teixeira da Silva Filho</i>	Dez/2001
36	Can Emerging Markets Float? Should They Inflation Target? <i>Barry Eichengreen</i>	Feb/2002
37	Monetary Policy in Brazil: Remarks on the Inflation Targeting Regime, Public Debt Management and Open Market Operations Luiz Fernando Figueiredo, Pedro Fachada and Sérgio Goldenstein	Mar/2002
38	Volatilidade Implícita e Antecipação de Eventos de Stress: um Teste para o Mercado Brasileiro <i>Frederico Pechir Gomes</i>	Mar/2002
39	Opções sobre Dólar Comercial e Expectativas a Respeito do Comportamento da Taxa de Câmbio <i>Paulo Castor de Castro</i>	Mar/2002
40	Speculative Attacks on Debts, Dollarization and Optimum Currency Areas <i>Aloisio Araujo and Márcia Leon</i>	Apr/2002
41	Mudanças de Regime no Câmbio Brasileiro Carlos Hamilton V. Araújo e Getúlio B. da Silveira Filho	Jun/2002
42	Modelo Estrutural com Setor Externo: Endogenização do Prêmio de Risco e do Câmbio Marcelo Kfoury Muinhos, Sérgio Afonso Lago Alves e Gil Riella	Jun/2002
43	The Effects of the Brazilian ADRs Program on Domestic Market Efficiency Benjamin Miranda Tabak and Eduardo José Araújo Lima	Jun/2002

44	Estrutura Competitiva, Produtividade Industrial e Liberação Comercial no Brasil Pedro Cavalcanti Ferreira e Osmani Teixeira de Carvalho Guillén	Jun/2002
45	Optimal Monetary Policy, Gains from Commitment, and Inflation Persistence <i>André Minella</i>	Aug/2002
46	The Determinants of Bank Interest Spread in Brazil Tarsila Segalla Afanasieff, Priscilla Maria Villa Lhacer and Márcio I. Nakane	Aug/2002
47	Indicadores Derivados de Agregados Monetários Fernando de Aquino Fonseca Neto e José Albuquerque Júnior	Set/2002
48	Should Government Smooth Exchange Rate Risk? Ilan Goldfajn and Marcos Antonio Silveira	Sep/2002
49	Desenvolvimento do Sistema Financeiro e Crescimento Econômico no Brasil: Evidências de Causalidade Orlando Carneiro de Matos	Set/2002
50	Macroeconomic Coordination and Inflation Targeting in a Two-Country Model	Sep/2002
	Eui Jung Chang, Marcelo Kfoury Muinhos and Joanílio Rodolpho Teixeira	
51	Credit Channel with Sovereign Credit Risk: an Empirical Test Victorio Yi Tson Chu	Sep/2002
52	Generalized Hyperbolic Distributions and Brazilian Data José Fajardo and Aquiles Farias	Sep/2002
53	Inflation Targeting in Brazil: Lessons and Challenges André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Nov/2002
54	Stock Returns and Volatility Benjamin Miranda Tabak and Solange Maria Guerra	Nov/2002
55	Componentes de Curto e Longo Prazo das Taxas de Juros no Brasil Carlos Hamilton Vasconcelos Araújo e Osmani Teixeira de Carvalho de Guillén	Nov/2002
56	Causality and Cointegration in Stock Markets: the Case of Latin America Benjamin Miranda Tabak and Eduardo José Araújo Lima	Dec/2002
57	As Leis de Falência: uma Abordagem Econômica Aloisio Araujo	Dez/2002
58	The Random Walk Hypothesis and the Behavior of Foreign Capital Portfolio Flows: the Brazilian Stock Market Case <i>Benjamin Miranda Tabak</i>	Dec/2002
59	Os Preços Administrados e a Inflação no Brasil Francisco Marcos R. Figueiredo e Thaís Porto Ferreira	Dez/2002
60	Delegated Portfolio Management Paulo Coutinho and Benjamin Miranda Tabak	Dec/2002

61	O Uso de Dados de Alta Freqüência na Estimação da Volatilidade e do Valor em Risco para o Ibovespa João Maurício de Souza Moreira e Eduardo Facó Lemgruber	Dez/2002
62	Taxa de Juros e Concentração Bancária no Brasil Eduardo Kiyoshi Tonooka e Sérgio Mikio Koyama	Fev/2003
63	Optimal Monetary Rules: the Case of Brazil Charles Lima de Almeida, Marco Aurélio Peres, Geraldo da Silva e Souza and Benjamin Miranda Tabak	Feb/2003
64	Medium-Size Macroeconomic Model for the Brazilian Economy Marcelo Kfoury Muinhos and Sergio Afonso Lago Alves	Feb/2003
65	On the Information Content of Oil Future Prices <i>Benjamin Miranda Tabak</i>	Feb/2003
66	A Taxa de Juros de Equilíbrio: uma Abordagem Múltipla Pedro Calhman de Miranda e Marcelo Kfoury Muinhos	Fev/2003
67	Avaliação de Métodos de Cálculo de Exigência de Capital para Risco de Mercado de Carteiras de Ações no Brasil Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Fev/2003
68	Real Balances in the Utility Function: Evidence for Brazil Leonardo Soriano de Alencar and Márcio I. Nakane	Feb/2003
69	r-filters: a Hodrick-Prescott Filter Generalization Fabio Araújo, Marta Baltar Moreira Areosa and José Alvaro Rodrigues Neto	Feb/2003
70	Monetary Policy Surprises and the Brazilian Term Structure of Interest Rates <i>Benjamin Miranda Tabak</i>	Feb/2003
71	On Shadow-Prices of Banks in Real-Time Gross Settlement Systems <i>Rodrigo Penaloza</i>	Apr/2003
72	O Prêmio pela Maturidade na Estrutura a Termo das Taxas de Juros Brasileiras <i>Ricardo Dias de Oliveira Brito, Angelo J. Mont'Alverne Duarte e Osmani</i> <i>Teixeira de C. Guillen</i>	Maio/2003
73	Análise de Componentes Principais de Dados Funcionais – uma Aplicação às Estruturas a Termo de Taxas de Juros Getúlio Borges da Silveira e Octavio Bessada	Maio/2003
74	Aplicação do Modelo de Black, Derman & Toy à Precificação de Opções Sobre Títulos de Renda Fixa Octavio Manuel Bessada Lion, Carlos Alberto Nunes Cosenza e César das Neves	Maio/2003
75	Brazil's Financial System: Resilience to Shocks, no Currency Substitution, but Struggling to Promote Growth <i>Ilan Goldfajn, Katherine Hennings and Helio Mori</i>	Jun/2003

76	Inflation Targeting in Emerging Market Economies Arminio Fraga, Ilan Goldfajn and André Minella	Jun/2003
77	Inflation Targeting in Brazil: Constructing Credibility under Exchange Rate Volatility André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Jul/2003
78	Contornando os Pressupostos de Black & Scholes: Aplicação do Modelo de Precificação de Opções de Duan no Mercado Brasileiro <i>Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo, Antonio</i> <i>Carlos Figueiredo, Eduardo Facó Lemgruber</i>	Out/2003
79	Inclusão do Decaimento Temporal na Metodologia Delta-Gama para o Cálculo do VaR de Carteiras Compradas em Opções no Brasil Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo, Eduardo Facó Lemgruber	Out/2003
80	Diferenças e Semelhanças entre Países da América Latina: uma Análise de <i>Markov Switching</i> para os Ciclos Econômicos de Brasil e Argentina Arnildo da Silva Correa	Out/2003
81	Bank Competition, Agency Costs and the Performance of the Monetary Policy <i>Leonardo Soriano de Alencar and Márcio I. Nakane</i>	Jan/2004
82	Carteiras de Opções: Avaliação de Metodologias de Exigência de Capital no Mercado Brasileiro Cláudio Henrique da Silveira Barbedo e Gustavo Silva Araújo	Mar/2004
83	Does Inflation Targeting Reduce Inflation? An Analysis for the OECD Industrial Countries <i>Thomas Y. Wu</i>	May/2004
84	Speculative Attacks on Debts and Optimum Currency Area: a Welfare Analysis Aloisio Araujo and Marcia Leon	May/2004
85	Risk Premia for Emerging Markets Bonds: Evidence from Brazilian Government Debt, 1996-2002 <i>André Soares Loureiro and Fernando de Holanda Barbosa</i>	May/2004
86	Identificação do Fator Estocástico de Descontos e Algumas Implicações sobre Testes de Modelos de Consumo Fabio Araujo e João Victor Issler	Maio/2004
87	Mercado de Crédito: uma Análise Econométrica dos Volumes de Crédito Total e Habitacional no Brasil Ana Carla Abrão Costa	Dez/2004
88	Ciclos Internacionais de Negócios: uma Análise de Mudança de Regime Markoviano para Brasil, Argentina e Estados Unidos Arnildo da Silva Correa e Ronald Otto Hillbrecht	Dez/2004
89	O Mercado de <i>Hedge</i> Cambial no Brasil: Reação das Instituições Financeiras a Intervenções do Banco Central Fernando N. de Oliveira	Dez/2004

90	Bank Privatization and Productivity: Evidence for Brazil Márcio I. Nakane and Daniela B. Weintraub	Dec/2004
91	Credit Risk Measurement and the Regulation of Bank Capital and Provision Requirements in Brazil – a Corporate Analysis <i>Ricardo Schechtman, Valéria Salomão Garcia, Sergio Mikio Koyama and</i> <i>Guilherme Cronemberger Parente</i>	Dec/2004
92	Steady-State Analysis of an Open Economy General Equilibrium Model for Brazil <i>Mirta Noemi Sataka Bugarin, Roberto de Goes Ellery Jr., Victor Gomes</i> <i>Silva, Marcelo Kfoury Muinhos</i>	Apr/2005
93	Avaliação de Modelos de Cálculo de Exigência de Capital para Risco Cambial Claudio H. da S. Barbedo, Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Abr/2005
94	Simulação Histórica Filtrada: Incorporação da Volatilidade ao Modelo Histórico de Cálculo de Risco para Ativos Não-Lineares Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo e Eduardo Facó Lemgruber	Abr/2005
95	Comment on Market Discipline and Monetary Policy by Carl Walsh <i>Maurício S. Bugarin and Fábia A. de Carvalho</i>	Apr/2005
96	O que É Estratégia: uma Abordagem Multiparadigmática para a Disciplina Anthero de Moraes Meirelles	Ago/2005
97	Finance and the Business Cycle: a Kalman Filter Approach with Markov Switching Ryan A. Compton and Jose Ricardo da Costa e Silva	Aug/2005
98	Capital Flows Cycle: Stylized Facts and Empirical Evidences for Emerging Market Economies <i>Helio Mori e Marcelo Kfoury Muinhos</i>	Aug/2005
99	Adequação das Medidas de Valor em Risco na Formulação da Exigência de Capital para Estratégias de Opções no Mercado Brasileiro Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo,e Eduardo Facó Lemgruber	Set/2005
100	Targets and Inflation Dynamics Sergio A. L. Alves and Waldyr D. Areosa	Oct/2005
101	Comparing Equilibrium Real Interest Rates: Different Approaches to Measure Brazilian Rates <i>Marcelo Kfoury Muinhos and Márcio I. Nakane</i>	Mar/2006
102	Judicial Risk and Credit Market Performance: Micro Evidence from Brazilian Payroll Loans Ana Carla A. Costa and João M. P. de Mello	Apr/2006
103	The Effect of Adverse Supply Shocks on Monetary Policy and Output Maria da Glória D. S. Araújo, Mirta Bugarin, Marcelo Kfoury Muinhos and Jose Ricardo C. Silva	Apr/2006

104	Extração de Informação de Opções Cambiais no Brasil <i>Eui Jung Chang e Benjamin Miranda Tabak</i>	Abr/2006
105	Representing Roommate's Preferences with Symmetric Utilities José Alvaro Rodrigues Neto	Apr/2006
106	Testing Nonlinearities Between Brazilian Exchange Rates and Inflation Volatilities <i>Cristiane R. Albuquerque and Marcelo Portugal</i>	May/2006
107	Demand for Bank Services and Market Power in Brazilian Banking Márcio I. Nakane, Leonardo S. Alencar and Fabio Kanczuk	Jun/2006
108	O Efeito da Consignação em Folha nas Taxas de Juros dos Empréstimos Pessoais Eduardo A. S. Rodrigues, Victorio Chu, Leonardo S. Alencar e Tony Takeda	Jun/2006
109	The Recent Brazilian Disinflation Process and Costs <i>Alexandre A. Tombini and Sergio A. Lago Alves</i>	Jun/2006
110	Fatores de Risco e o Spread Bancário no Brasil Fernando G. Bignotto e Eduardo Augusto de Souza Rodrigues	Jul/2006
111	Avaliação de Modelos de Exigência de Capital para Risco de Mercado do Cupom Cambial Alan Cosme Rodrigues da Silva, João Maurício de Souza Moreira e Myrian Beatriz Eiras das Neves	Jul/2006
112	Interdependence and Contagion: an Analysis of Information Transmission in Latin America's Stock Markets <i>Angelo Marsiglia Fasolo</i>	Jul/2006
113	Investigação da Memória de Longo Prazo da Taxa de Câmbio no Brasil Sergio Rubens Stancato de Souza, Benjamin Miranda Tabak e Daniel O. Cajueiro	Ago/2006
114	The Inequality Channel of Monetary Transmission Marta Areosa and Waldyr Areosa	Aug/2006
115	Myopic Loss Aversion and House-Money Effect Overseas: an Experimental Approach <i>José L. B. Fernandes, Juan Ignacio Peña and Benjamin M. Tabak</i>	Sep/2006
116	Out-Of-The-Money Monte Carlo Simulation Option Pricing: the Join Use of Importance Sampling and Descriptive Sampling <i>Jaqueline Terra Moura Marins, Eduardo Saliby and Joséte Florencio dos</i> <i>Santos</i>	Sep/2006
117	An Analysis of Off-Site Supervision of Banks' Profitability, Risk and Capital Adequacy: a Portfolio Simulation Approach Applied to Brazilian Banks Theodore M. Barnhill, Marcos R. Souto and Benjamin M. Tabak	Sep/2006
118	Contagion, Bankruptcy and Social Welfare Analysis in a Financial Economy with Risk Regulation Constraint <i>Aloísio P. Araújo and José Valentim M. Vicente</i>	Oct/2006

119	A Central de Risco de Crédito no Brasil: uma Análise de Utilidade de Informação Ricardo Schechtman	Out/2006
120	Forecasting Interest Rates: an Application for Brazil Eduardo J. A. Lima, Felipe Luduvice and Benjamin M. Tabak	Oct/2006
121	The Role of Consumer's Risk Aversion on Price Rigidity Sergio A. Lago Alves and Mirta N. S. Bugarin	Nov/2006
122	Nonlinear Mechanisms of the Exchange Rate Pass-Through: a Phillips Curve Model With Threshold for Brazil <i>Arnildo da Silva Correa and André Minella</i>	Nov/2006
123	A Neoclassical Analysis of the Brazilian "Lost-Decades" Flávia Mourão Graminho	Nov/2006
124	The Dynamic Relations between Stock Prices and Exchange Rates: Evidence for Brazil <i>Benjamin M. Tabak</i>	Nov/2006
125	Herding Behavior by Equity Foreign Investors on Emerging Markets Barbara Alemanni and José Renato Haas Ornelas	Dec/2006
126	Risk Premium: Insights over the Threshold José L. B. Fernandes, Augusto Hasman and Juan Ignacio Peña	Dec/2006
127	Uma Investigação Baseada em Reamostragem sobre Requerimentos de Capital para Risco de Crédito no Brasil Ricardo Schechtman	Dec/2006
128	Term Structure Movements Implicit in Option Prices <i>Caio Ibsen R. Almeida and José Valentim M. Vicente</i>	Dec/2006
129	Brazil: Taming Inflation Expectations Afonso S. Bevilaqua, Mário Mesquita and André Minella	Jan/2007
130	The Role of Banks in the Brazilian Interbank Market: Does Bank Type Matter? <i>Daniel O. Cajueiro and Benjamin M. Tabak</i>	Jan/2007
131	Long-Range Dependence in Exchange Rates: the Case of the European Monetary System Sergio Rubens Stancato de Souza, Benjamin M. Tabak and Daniel O. Cajueiro	Mar/2007
132	Credit Risk Monte Carlo Simulation Using Simplified Creditmetrics' Model: the Joint Use of Importance Sampling and Descriptive Sampling <i>Jaqueline Terra Moura Marins and Eduardo Saliby</i>	Mar/2007
133	A New Proposal for Collection and Generation of Information on Financial Institutions' Risk: the Case of Derivatives <i>Gilneu F. A. Vivan and Benjamin M. Tabak</i>	Mar/2007
134	Amostragem Descritiva no Apreçamento de Opções Européias através de Simulação Monte Carlo: o Efeito da Dimensionalidade e da Probabilidade de Exercício no Ganho de Precisão Eduardo Saliby, Sergio Luiz Medeiros Proença de Gouvêa e Jaqueline Terra Moura Marins	Abr/2007

135	Evaluation of Default Risk for the Brazilian Banking Sector <i>Marcelo Y. Takami and Benjamin M. Tabak</i>	May/2007
136	Identifying Volatility Risk Premium from Fixed Income Asian Options Caio Ibsen R. Almeida and José Valentim M. Vicente	May/2007
137	Monetary Policy Design under Competing Models of Inflation Persistence Solange Gouvea e Abhijit Sen Gupta	May/2007
138	Forecasting Exchange Rate Density Using Parametric Models: the Case of Brazil Marcos M. Abe, Eui J. Chang and Benjamin M. Tabak	May/2007
139	Selection of Optimal Lag Length inCointegrated VAR Models with Weak Form of Common Cyclical Features Carlos Enrique Carrasco Gutiérrez, Reinaldo Castro Souza and Osmani Teixeira de Carvalho Guillén	Jun/2007
140	Inflation Targeting, Credibility and Confidence Crises Rafael Santos and Aloísio Araújo	Aug/2007
141	Forecasting Bonds Yields in the Brazilian Fixed income Market Jose Vicente and Benjamin M. Tabak	Aug/2007
142	Crises Análise da Coerência de Medidas de Risco no Mercado Brasileiro de Ações e Desenvolvimento de uma Metodologia Híbrida para o Expected Shortfall Alan Cosme Rodrigues da Silva, Eduardo Facó Lemgruber, José Alberto Rebello Baranowski e Renato da Silva Carvalho	Ago/2007
143	Price Rigidity in Brazil: Evidence from CPI Micro Data Solange Gouvea	Sep/2007
144	The Effect of Bid-Ask Prices on Brazilian Options Implied Volatility: a Case Study of Telemar Call Options <i>Claudio Henrique da Silveira Barbedo and Eduardo Facó Lemgruber</i>	Oct/2007
145	The Stability-Concentration Relationship in the Brazilian Banking System <i>Benjamin Miranda Tabak, Solange Maria Guerra, Eduardo José Araújo</i> <i>Lima and Eui Jung Chang</i>	Oct/2007
146	Movimentos da Estrutura a Termo e Critérios de Minimização do Erro de Previsão em um Modelo Paramétrico Exponencial Caio Almeida, Romeu Gomes, André Leite e José Vicente	Out/2007